Author's response to reviews

Title: Physical activity and health related quality of life

Authors:

Nana K Anokye (nana.anokye@brunel.ac.uk)
Paul Trueman (Paul.Trueman@brunel.ac.uk)
Colin Green (colin.green@pms.ac.uk)
Toby G Pavey (tgpavey@pcmd.ac.uk)
Rod S Taylor (rod.taylor@pms.ac.uk)

Version: 2 Date: 24 February 2012

Author's response to reviews:

Dear Editor

As per the requested revision, I have done the following:

(a) added aims of the study to the background section of the abstract
(b) I have indicated in the main text that the data (first sentence in data section) for analysis is openly available,
(c) I have added a text on author contributions after competing interests

Many thanks,
Nana Anokye